Author's response to reviews

Title: A Community Participator Approach to Improving Health in a Hispanic Population

Authors:

Michael F Dulin (michael.dulin@carolinashealthcare.org)
Heather A Smith (heatsmit@uncc.org)
Hazel Tapp (hazel.tapp@carolinashealthcare.org)
Brisa Urquieta De Hernandez (Brisa.Hernandez@carolinashealthcare.org)
Owen J Furuseth (ojfuruse@uncc.edu)

Version: 2 Date: 29 December 2010

Author's response to reviews:

Thank you very much, we really appreciate the opportunity to make revisions to our submission. Please find included our revised manuscript. We have now included the abstract on page 2 of the manuscript, the ethics statement in the methods and a discussion section at the end of the manuscript. I would like to submit the Table in additional material as a figure rather than supplemental material. Additionally, we have included the legend to figure 4 as an additional Figure. Again, we do sincerely thank you all for your kind consideration of our manuscript.